Author's response to reviews

Title: Relationship between Organizational Culture, Leadership Behavior and Job Satisfaction

Authors:

    Yafang Tsai (avon611@gmail.com)

Version: 6 Date: 24 April 2011

Author's response to reviews: see over
Dear Editor in Chief,

Thank you for accepting this paper to publish on BMC Health Services Research. The author already had checked the format of the paper. Because this is one-author work, I don’t write the “Competing interests” and “Authors contributions” in the paper.

Sincerely,

Yafang Tsai

Associate Professor, Department of Health Policy and Management, Chung Shan Medical University. Department of Medical Management, Chung Shan Medical University Hospital, Taichung City, Taiwan.
No.82, Sec. 2, Chongsing Rd. Taichung City 404, Taiwan.
Tel: 886- 4-24730022 ext 12127
Email: avon611@gmail.com, avon611@yahoo.com.tw